Author’s response to reviews

Title: The Checkpointkinase 2 (CHK2) 1100delC Germ Line Mutation is not associated with the Development of Squamous Cell Carcinoma of the Head and Neck (SCCHN)

Authors:

Kathrin Scheckenbach Dr (scheckenbach@med.uni-duesseldorf.de)
Galatia Papadopoulou (galatia-papadopoulou@gmx.de)
Thomas K Hoffmann Prof (thomas.hoffmann@uk-essen.de)
Adam Chaker Dr (adam@chaker.de)
Henning Bier Prof (H.Bier@lrz.tum.de)
Jörg Schipper Prof (joerg.schipper@med.uni-duesseldorf.de)
Vera Balz Dr (balz@uni-duesseldorf.de)
Martin Wagenmann Dr (martin.Wagenmann@uni-duesseldorf.de)

Version: 2 Date: 28 November 2010

Author’s response to reviews: see over
Dear editors,

Thank you very much for comments concerning our manuscript with the title:

**The Checkpointkinase 2 (CHK2) 1100delC Germ Line Mutation is not associated with the Development of Squamous Cell Carcinoma of the Head and Neck (SCCHN)**

for consideration as a publication for "Journal of Negative Results in Biomedicine" in the category of "Research Article".

We revised the manuscript according to your suggestions and hereby send it you again. We are optimistic that we were able to address all points that were made and hope that you will publish our paper in your journal. In order to make the changes to the manuscript more visible we uploaded an additional Word-document with markups under „Additional material files“ that is identical to the revised manuscript except for the markups that highlight the modifications.

Please don’t hesitate to contact me again if there are any remaining questions.

Yours, sincerely

Kathrin Scheckenbach